Author's response to reviews

Title: Learning to Prescribe - Pharmacists' Experiences of Supplementary Prescribing Training

Authors:

Richard J Cooper (richard.cooper@nottingham.ac.uk)
Joanne Lymn (joanne.lymn@nottingham.ac.uk)
Claire Anderson (claire.anderson@nottingham.ac.uk)
Tony Avery (a.avery@nottingham.ac.uk)
Paul Bissell (P.bissell@sheffield.ac.uk)
Louise Guillaume (l.r.guillaume@sheffield.ac.uk)
Allen Hutchinson (allen.hutchinson@sheffield.ac.uk)
Elizabeth Murphy (e.murphy@nottingham.ac.uk)
Julie Ratcliffe (julie.ratcliffe@unisa.edu.au)
Paul Ward (Paul.ward@flinders.edu.au)

Version: 3 Date: 18 November 2008

Author's response to reviews:

Dear Editors,

The manuscript has been further revised as suggested and now includes a corrected "P<0.001" value and a reduction in the discussion section of around 300 words (half a page) as suggested by one of the reviewers. The manuscript and in particular the referencing have been re-checked and amended to conform to the correct format/style.

I hope that these changes will enable the paper to be published,

Regards,

Richard Cooper